Author’s response to reviews

Title: Sero-prevalence and risk factors for hepatitis B infection among health care workers in a tertiary hospital in Uganda

Authors:

abdhalah K Ziraba (akziraba@yahoo.com)
Josephine Bwogi (josephinebwn@yahoo.co.uk)
Alice Namale (asnamal@hotmail.com)
Caroline W Wainaina (carowangoi@yahoo.com)
Harriet Mayanja-Kizza (hmk@chs.mak.ac.ug)

Version: 8 Date: 21 June 2010

Author’s response to reviews:

June, 21st, 2010

Editor-in-Chief,
BMC Infectious Diseases Journal

MS: 1630543822968744- Sero-prevalence and risk factors for hepatitis B infection among health care workers in a tertiary hospital in Uganda

Dear Editor,

Thank you for accepting to publish our manuscript. In response to your email of June 18th, 2010, we are pleased to submit a revised version of the manuscript. The key highlights are as follows:

i) We have reconciled the authors names in the manuscript to appear as those in the BMC system

ii) Authors contribution: We have added “All authors read and approved the final manuscript”

iii) All track change marks and highlights have been removed

iv) Table placement text has been removed

v) The manuscript has been proof read and a few minors edits carried out.

Kind regards

Abdhalah Ziraba (Corresponding author)